Author's response to reviews

Title: 4th Stage Transvaginal Omental Herniation during repeat VBAC complicated by shoulder dystocia: a unique presentation of uterine rupture

Authors:

Aruyaru Stanley Mwenda (arumwesta@gmail.com)

Version: 2 Date: 30 December 2012

Author's response to reviews:

Dear editor,

I have revised the initial manuscript and now it includes the following:

1. A title page with the author’s affiliation and contact details
2. A consent section indicating that informed consent was given for the publication of this article and accompanying images
3. A section on competing interests has been included

kindly receive the revision uploaded